

Distinct Central and Eastern Pacific El Niño Influence on Antarctic Surface Mass Balance

J M A Macha, A N Mackintosh, F S Mccormack, B J Henley, H V Mcgregor,

C T van Dalum, A. Purich

► To cite this version:

J M A Macha, A N Mackintosh, F S Mccormack, B J Henley, H V Mcgregor, et al.. Distinct Central and Eastern Pacific El Niño Influence on Antarctic Surface Mass Balance. Geophysical Research Letters, 2024, 51 (11), 10.1029/2024gl109423. hal-04670442

HAL Id: hal-04670442 https://hal.science/hal-04670442v1

Submitted on 12 Aug 2024

HAL is a multi-disciplinary open access archive for the deposit and dissemination of scientific research documents, whether they are published or not. The documents may come from teaching and research institutions in France or abroad, or from public or private research centers. L'archive ouverte pluridisciplinaire **HAL**, est destinée au dépôt et à la diffusion de documents scientifiques de niveau recherche, publiés ou non, émanant des établissements d'enseignement et de recherche français ou étrangers, des laboratoires publics ou privés.

ŗ



Geophysical Research Letters[•]

RESEARCH LETTER

10.1029/2024GL109423

Key Points:

- Central and Eastern Pacific El Niño events have distinctly different seasonal impacts on Antarctic surface climate
- Central Pacific El Niño accumulation anomalies develop earlier than Eastern Pacific El Niño, and persist through austral winter and spring
- Central Pacific El Niño impacts on West Antarctic accumulation are significantly greater than those of Eastern Pacific El Niño events

Supporting Information:

Supporting Information may be found in the online version of this article.

Correspondence to:

J. M. A. Macha, jessica.macha@monash.edu

Citation:

Macha, J. M. A., Mackintosh, A. N., McCormack, F. S., Henley, B. J., McGregor, H. V., van Dalum, C. T., & Purich, A. (2024). Distinct Central and Eastern Pacific El Niño influence on Antarctic surface mass balance. *Geophysical Research Letters*, 51, e2024GL109423. https://doi.org/10.1029/ 2024GL109423

Received 21 MAR 2024 Accepted 16 MAY 2024

Author Contributions:

Conceptualization: J. M. A. Macha Data curation: C. T. van Dalum Formal analysis: J. M. A. Macha Investigation: J. M. A. Macha Methodology: J. M. A. Macha Project administration: J. M. A. Macha Resources: J. M. A. Macha Supervision: A. N. Mackintosh, F. S. McCormack, B. J. Henley, H. V. McGregor Validation: J. M. A. Macha Visualization: J. M. A. Macha

© 2024. The Author(s). This is an open access article under the terms of the Creative Commons Attribution-NonCommercial-NoDerivs License, which permits use and distribution in any medium, provided the original work is properly cited, the use is non-commercial and no modifications or adaptations are made.

Distinct Central and Eastern Pacific El Niño Influence on Antarctic Surface Mass Balance

J. M. A. Macha¹, A. N. Mackintosh¹, F. S. McCormack¹, B. J. Henley^{2,3,4}, H. V. McGregor^{3,4}, C. T. van Dalum⁵, and A. Purich¹

¹School of Earth, Atmosphere and Environment, Securing Antarctica's Environmental Future, Monash University, Clayton, VIC, Australia, ²School of Agriculture, Food & Ecosystem Sciences, University of Melbourne, Burnley, VIC, Australia, ³School of Earth, Atmosphere and Life Sciences, Securing Antarctica's Environmental Future, University of Wollongong, Wollongong, NSW, Australia, ⁴School of Earth, Atmosphere and Life Sciences, Environmental Futures, University of Wollongong, Wollongong, NSW, Australia, ⁵Utrecht University, Institute for Marine and Atmospheric Research Utrecht, Utrecht, The Netherlands

Abstract The El Niño-Southern Oscillation causes anomalous atmospheric circulation, temperature and precipitation across southern polar latitudes, but the influence of Central and Eastern Pacific El Niño events on Antarctic surface mass balance and snow accumulation has not yet been assessed. Here, we use reanalysis and reanalysis-forced regional climate model output and find that Central Pacific El Niño results in significantly increased snow accumulation in the western Ross Sea sector and significantly decreased snow accumulation in the Amundsen Sea sector. Eastern Pacific El Niño is associated with similar but weaker patterns, with some regional exceptions. In some areas, like Dronning Maud Land, or the Wilkes Subglacial Basin, the effect of El Niño on snow accumulation changes from increased to reduced accumulation depending on the type of El Niño. Our results show that projecting El Niño types is important for constraining future changes in Antarctic surface mass balance.

Plain Language Summary The El Niño-Southern Oscillation influences the climate in Antarctica with impacts on temperature, rainfall and snow accumulation. However, the effect of two different types of El Niño events, Central Pacific and Eastern Pacific El Niño, on the patterns of snow accumulation in Antarctica has not yet been studied. We show from reanalysis and reanalysis-forced regional climate model output that Central and Eastern Pacific El Niño events have distinct effects on Antarctic snow accumulation patterns. During Central Pacific El Niño events, there is an increase in snow accumulation in the western Ross Sea sector, and decrease in the Amundsen Sea region. Eastern Pacific El Niño events cause similar regional effects, but to a lesser extent. In some areas, like Dronning Maud Land or the Wilkes Subglacial Basin, the effect of El Niño on snow accumulation changes from increased to reduced accumulation depending on the type of El Niño. Our findings emphasize the importance of understanding projected changes in El Niño types, as these will impact the future of Antarctica's snow accumulation.

1. Introduction

The Antarctic Ice Sheet (AIS) has been losing mass for several decades due to regional ocean-driven increases in basal melting of ice shelves, iceberg calving and associated dynamic changes in ice discharge (Frieler et al., 2015; Fox-Kemper et al., 2021; Hanna et al., 2020, 2024; J.-Y. Lee et al., 2021). This ocean-driven mass loss is counteracted by mass gain in some regions of Antarctica, driven by processes underlying surface mass balance (SMB), that is, the sum of accumulation and ablation (melting) processes at the ice sheet surface (Agosta et al., 2013; Mottram et al., 2021). Snow accumulation, the major control on SMB in Antarctica, has been increasing over this period, although with significant spatial and temporal variability (Hanna et al., 2020; Noël et al., 2023). Understanding and projecting the balance between mass loss due to basal melting and calving and mass gain due to snow accumulation is essential for understanding the Antarctic contribution to global sea level rise. Here, we investigate the climatological drivers of Antarctic SMB variability with a focus on the El Niño-Southern Oscillation (ENSO).

Snow accumulation in Antarctica is spatially and temporally variable, and accurate projections have been hampered by a paucity of long-term high-resolution observations (Frieler et al., 2015; Fox-Kemper et al., 2021; J.-Y. Lee et al., 2021). Three priorities for advancing research on the drivers of Antarctic accumulation change are:



Writing – review & editing: J. M. A. Macha, A. N. Mackintosh.

C. T. van Dalum, A. Purich

F. S. McCormack, H. V. McGregor,

(a) understanding how large-scale atmospheric processes influence the SMB of the AIS; (b) understanding how Antarctic climate variability is connected to the mid-latitudes and tropics; and (c) determining the controls on regional patterns of Antarctic climate (Kennicutt et al., 2014, 2015). An improved characterization of how climate variability impacts Antarctic accumulation is important because regional accumulation increases, for example, in Dronning Maud Land in East Antarctica (Diener et al., 2021), have the potential to offset (on a continental scale) some of the projected ocean-driven mass losses in a warming climate (Fox-Kemper et al., 2021; Kittel et al., 2021; Seroussi et al., 2020).

There are several known climatological drivers of Antarctic surface climate, including the Southern Annular Mode (SAM) (Dätwyler et al., 2018; Fogt & Marshall, 2020; Medley & Thomas, 2019; Saunderson et al., 2024) and the Atlantic Multi-decadal Oscillation (Li et al., 2014, 2021). Here, we focus on ENSO, which is the leading mode globally of interannual climate variability, and exerts strong influences in Antarctica (Cullather et al., 1996; J. King & Turner, 1997; Ribera & Mann, 2003). El Niño impacts Antarctica by causing a weakening and shift in the location of the Amundsen Sea Low (ASL), through an atmospheric Rossby wave teleconnection (Karamperidou et al., 2020; Turner, 2004), with consequent changes in atmospheric circulation and precipitation patterns over Antarctica (Bodart & Bingham, 2019; Bromwich et al., 2004; Guo et al., 2004; Sun et al., 2013). El Niño events have also been shown to impact Antarctic temperature (T. Lee et al., 2010; Wilson et al., 2014), sea ice (C. Zhang, Li, & Li, 2021), surface melt (Clem et al., 2022), and ice shelf thickness (Paolo et al., 2018).

El Niño events fall into different types, defined by sea surface temperature (SST) anomalies in the tropical Pacific, with peak anomalies in either the central equatorial Pacific—denoted "CP El Niño"—or the eastern equatorial Pacific—denoted "EP El Niño"—(Ashok et al., 2007; Kug et al., 2009; Song et al., 2011). Distinct SST anomalies associated with CP and EP El Niño affect the location of maximum deep atmospheric convection in the tropical Pacific, causing the source location of Rossby wave trains to differ, occurring 20°–30° west during CP El Niño compared with EP El Niño (Chen et al., 2023; Clem et al., 2017; H.-J. Lee & Jin, 2021; C. Zhang, Li, & Li, 2021). This results in different Rossby wave propagation pathways to Antarctica resulting from CP and EP El Niño (Capotondi et al., 2015; Chen et al., 2023; Clem et al., 2017; H.-J. Lee & Jin, 2021; C. Zhang et al., 2021). C. Zhang, Li, and Li (2021) and Clem et al. (2017) show that the CP El Niño Rossby wave train results in a weakening and westward latitudinal shift in the ASL compared to EP El Niño.

Previous studies have identified ENSO signals in AIS SMB changes and elevation changes (Mémin et al., 2015; Kaitheri et al., 2021; Pfeffer et al., 2022; Zhan et al., 2021; C. Zhang, Li, &, 2021). However, no studies to date have considered the impacts of different El Niño types on SMB in Antarctica (M. King et al., 2023; Sasgen et al., 2010). Here, we explore the temperature, precipitation and SMB changes over a 40-year period from 1979 to 2018 associated with different El Niño types (Figure S1 in Supporting Information S1).

2. Data Sets and Methods

2.1. Regional Climate Model Data: RACMO2.3p3

We assess climate variability over the Antarctic continent using output from the Regional Atmospheric Climate Model version 2.3p3 (RACMO2.3p3) (van Dalum et al., 2021, 2022). RACMO2.3p3 covers the Antarctic domain at 27 km resolution and is forced by 3-hourly global output from the European Center for Medium-Range Weather Forecasts atmospheric reanalysis (ERA5) for 1979–2018 (Hersbach et al., 2020; van Dalum et al., 2022). ERA5 is a global reanalysis, which assimilates observations and models to optimally produce data sets of global atmosphere, land surface and ocean variables since 1940 (Hersbach et al., 2020).

Antarctic surface elevation and drainage basins including catchment and regional names used in this study, are shown in Figure 1. We use monthly SMB, total precipitation and 2-m air temperature from RACMO2.3p3 for 1979–2018 (van Dalum et al., 2021, Figure S1 in Supporting Information S1). While 1979–1984 is considered a spin up period in RACMO2.3p3, we show that including or excluding the period 1979–1984 has little impact on our results (see Text S1, Figure S2–S3 in Supporting Information S1). This suggests that this spin-up period is still representative of Antarctic climatology and we therefore utilize the full 40-year output period in our study, as a longer period is beneficial for examining interannual variability. Seasonal averages are calculated over: December–February (DJF), March–May (MAM), June–August (JJA), and September–November (SON). This study focuses on austral winter (JJA) and austral spring (SON), when the ENSO-Antarctic teleconnection impacts



Geophysical Research Letters



Figure 1. Antarctica surface elevation (contours; meters above sea level) from Liu et al. (2001), with individual drainage basins (thick black lines) and drainage basins subregions (thin black lines) from Rignot, Mouginot, and Scheuchl (2011) and Rignot, Velicogna, et al. (2011). Inset map with subregions defined by Rignot, Mouginot, and Scheuchl (2011). Amundsen Sea Low shown schematically.

are strongest and the ASL is located furthest southwest (Jin & Kirtman, 2009; Li et al., 2021; Renwick & Revell, 1999; Yiu & Maycock, 2020).

2.2. Reanalysis Data

We examine 0.25° resolution Mean Sea Level Pressure (MSLP) and 10-m wind vectors from ERA5 for the entire region south of 40°S (Hersbach et al., 2020). ERA5 covers a larger geographical extent than RACMO2.3p3, enabling analysis of atmospheric variables across the southern mid-latitudes (Hersbach et al., 2020; van Dalum et al., 2022). We use monthly and 3-month seasonal averages from 1979 to 2018 to match the period covered by RACMO2.3p3 data.

2.3. El Niño Indices

El Niño types are defined according to the location of peak SST anomalies in the tropical Pacific (Ashok et al., 2007; Kug et al., 2009). Central Pacific (CP) El Niño events and Eastern Pacific (EP) El Niño events are classified according to the Ren and Jin (2011) N_{CP} and N_{EP} indices respectively:

$$N_{EP} = N_3 - \alpha N_4 \tag{1}$$

$$N_{CP} = N_4 - \alpha N_3$$

where

 $\alpha = \begin{cases} \frac{2}{5}, N_3 N_4 > 0, \\ 0, \text{ otherwise.} \end{cases}$

1

Here, N_3 is the Niño-3 index, which is the SST anomaly averaged over the regions 5°N-5°S and 150°-90°W, and N_4 is the Niño-4 index, which is the SST anomaly averaged over the regions 5°N–5°S and 160°E–150°W (Ren & Jin, 2011). Niño-3 and Niño-4 indices are sourced from NOAA (Rayner, 2003), based on the Hadley Center Sea Ice and Sea Surface Temperature data set (HadISST). We use 3-month seasonal averages from 1979 to 2018 of CP and EP El Niño indices (Equations 1 and 2). Between 1979 and 2018 five EP El Niño events (1982/83, 1991/92, 1997/98, 2006/07, 2015/16) and six CP El Niño events (1994/95, 2002/03, 2004/05, 2009/10, 2014/15, 2018/19) occur (extended data from Ren and Jin (2011)). We disregard the 1986/1988 El Niño event, which began as an EP El Niño but then matured into a CP El Niño (Ren & Jin, 2011). To maximize the length of the timeseries for our analysis, we exclude all negative values from each index (i.e., apply a threshold criterion of 0), which removes all La Niña (any flavor) years. This results in 22 CP El Niño data points and 15 EP El Niño events over the 40-year period.

2.4. Statistical Analyses

Power spectra are computed for total continental Antarctica, West Antarctica, and in the Antarctic Peninsula separately, using Welch overlapping segment averaging on mean detrended RACMO2.3p3 SMB, an oversampling value of two, and a segment value of two (Mudelsee, 2010, see Text S2, Figure S4 in Supporting Information S1). Signal peaks at ~ 12 , ~ 5.8 , and ~ 2.4 years occur at the continental scale SMB (Figure S4a in Supporting Information S1), and at a regional scale in West Antarctica (Figure S4b in Supporting Information S1) and the Antarctic Peninsula (Figure S4c in Supporting Information S1), confirming that interannual variability dominates monthly Antarctic SMB.

We perform statistical analyses between Antarctic surface climate and different El Niño types. We do not perform Rossby wave dynamics analysis, as previous studies have established the changes associated with Rossby wave propagation to the southern high latitudes during CP and EP El Niño events (H.-J. Chen et al., 2023; Lee & Jin, 2021; Li et al., 2021). Here, we focus on statistical analyses between Antarctic surface climate and different El Niño types.

ENSO-Antarctic teleconnection impacts are strongest in JJA and SON when the subtropical jet is strongest, enabling the propagation of Rossby waves from the tropics to the poles; this is also when the ASL is shifted southwest (Jin & Kirtman, 2009; Li et al., 2021; Renwick & Revell, 1999; Yiu & Maycock, 2020). Therefore, we focus on JJA and SON results (Figures 2-4); other seasonal and lagged results are included in supplementary Figures S6–S9 in Supporting Information S1. We regress seasonal CP and EP El Niño indices (Equations 1 and 2) against seasonal 2-m air temperature, precipitation, SMB, MSLP and 10-m wind speed, relative to the 1979-2018 averages of these variables. This shows the seasonal changes in these variables associated with a 1°C strength CP or EP El Niño event, with tropical Pacific SST anomalies of 1°C (see Text S3 in Supporting Information S1, Figure 2). We also undertake composite analysis on seasonal CP and EP El Niño indices (Equations 1 and 2) against seasonal JJA and SON 2-m air temperature, precipitation, SMB, MSLP and 10-m wind speed for 1979-2018, to confirm robustness of regression results (see Text S4 in Supporting Information S1, Figures 3b, 3e–3f, 3i–3i, Figure S5 in Supporting Information S1). All fields are linearly detrended prior to calculating regressions and composites, to remove any long term trend over the timeseries. Statistical significance is established using a Students t test at the 5% confidence level. All results presented are statistically significant unless otherwise

19448007, 2024, 11, Downlo

ided from https://

com/doi/10.1029/2024GL109423 by Cochrane France

(2)



Figure 2. Modeled ENSO-Antarctic climate regressions in JJA (a–f) and SON(g-l): (a, d, g, j) 2-m air temperature (temp; from RACMO2.3p3) and mean sea level pressure (MSLP (gray; 2 hPa increments, (dashed lines are negative) from ERA5) (b, e, h, k) precipitation (Precip; from RACMO2.3p3), and (c, f, i, l) SMB (from RACMO2.3p3). Temperature, MSLP, precipitation and SMB anomalies represent seasonal changes in temperature, MSLP, precipitation and SMB when these variables are regressed against the JJA and SON CP or EP El Niño index relative to the 1979–2018 averages of these variables. These changes in temperature, MSLP, precipitation and SMB when these variables. These changes in temperature, MSLP, precipitation and SMB are associated with a 1°C strength CP and EP El Niño event when tropical Pacific SST anomalies are 1°C. Fields are linearly detrended prior to regressions onto the: N_{CP} index (first column, a, c, e) and N_{EP} index (second column, b, d, f) for 1979–2018 from Ren and Jin (2011) (methods). Stippling indicates statistical significance at the 5% confidence level using a two-tailed Students' *t* test.

indicated. Auto-correlation is accounted for by utilizing seasonal averages (Mudelsee, 2010, see Text S5 in Supporting Information S1).

3. Results

3.1. CP and EP El Niño Impacts on Regional Climate and SMB

3.1.1. CP El Niño

Figure 2 shows the seasonal changes in temperature, MSLP, precipitation and SMB when these variables are regressed against the JJA and SON CP or EP El Niño index relative to the 1979–2018 averages of these variables. These anomalies show changes in temperature, MSLP, precipitation and SMB which are associated with a 1°C strength CP and EP El Niño event when tropical Pacific SST anomalies are 1°C. During CP El Niño, in both JJA

5 of 12

and SON, positive temperature anomalies are widespread across the Amundsen and Ross Seas, and negative anomalies persist across the Antarctic Peninsula (Figures 1 and 2a, Figure S5a in Supporting Information S1). We refer to these spatial patterns as "dipoles" (Figure 2a). In JJA negative temperature anomalies also occur across the Weddell Sea (Figures 1 and 2a) whilst in SON, positive temperature anomalies occur in the Weddell Sea and Ronne-Filchner catchment (Figure 2g, Figure S5g in Supporting Information S1).

A dipole pattern also occurs in precipitation and SMB anomalies during CP El Niño in JJA, with increases in the Ross West sector and decreases across the Amundsen Sea sector (Figures 2b and 2c, 2h and 2i). During SON similar precipitation and SMB increases occur in the Ross West sector, but decreases across the Amundsen Sea sector are not significant (Figures 2h and 2i, Figures S5h and S5i in Supporting Information S1).

3.1.2. EP El Niño

During EP El Niño, there is decreased temperature in JJA across West Antarctica relative to CP El Niño, but there is no significant temperature change over the Peninsula (Figure 2d, Figure S5d in Supporting Information S1). In SON, during EP El Niño, a positive temperature anomaly occurs across the Amundsen and Ross Sea sectors, whilst a negative temperature anomaly extends over both the Antarctic Peninsula and Weddell Sea (Figure 2j). Temperature anomalies in West Antarctica during EP El Niño in SON are similar to those during CP El Niño in JJA and SON (Figure 2a, 2g, and 2j, Figure S5j in Supporting Information S1).

The magnitude of the anomalies in atmospheric circulation and thus in the resulting precipitation and SMB is lower during EP El Niño than CP El Niño—highlighting that CP El Niño has a much greater impact on Antarctic SMB than EP El Niño (Figures 2a–2f). In contrast with CP El Niño, EP El Niño events have very little impact on precipitation and SMB fields during JJA (Figures 2e and 2f). However, during SON, there are increases in precipitation and SMB in the Ross West basin and decreases in precipitation and SMB in the Peninsula and Amundsen Sea region during EP El Niño events—a similar pattern to the changes occurring during CP El Niño events, but of a lesser magnitude and less widespread (Figures 2h and 2i, 2k and 2l).

In summary, our results reveal distinctly different spatial signatures in atmospheric variables and SMB during CP and EP El Niño events, with larger magnitude and more widespread changes occurring during CP El Niño compared to EP El Niño (Figure 2, Figure S5 in Supporting Information S1). Furthermore, we show different spatial CP and EP El Niño impacts in JJA and SON, revealing that CP and EP El Nino do not have the same teleconnective impacts during the same seasons (Figures 2 and 3, Figure S5 in Supporting Information S1). Strong teleconnective relationships occur between the tropical Pacific and Antarctica during both JJA and SON, and weaker relationships occur during DJF and MAM (Figure 2, Figures S6 and S7 in Supporting Information S1). East Antarctic impacts generally differ between drainage basins and are mostly not statistically significant; see Texts S6–S7 in Supporting Information S1 for further discussion of the East Antarctic impacts.

3.2. Atmospheric Drivers of El Niño SMB Impacts in West Antarctica

During CP El Niño, positive MSLP changes occur over the West Antarctic coastline in the Amundsen Sea, with a consequent weakening and westwards shift in the ASL (Figures 3c–3f). This results in strong easterly, moistureladen onshore winds flowing into West Antarctica from the Ross Sea sector (Figures S13a, S13c, S13e, and S13g in Supporting Information S1). Orographic uplift results in precipitation increases in the region (Figures 2b and 2h), which drive SMB gain (Figures 3a–3f). Simultaneously, over the Amundsen Sea sector and the Antarctic Peninsula, downslope winds (Figures S13a, S13c, S13e, and S13g in Supporting Information S1) are associated with reduced precipitation and SMB decreases (Figures 3a–3f).

The ASL only slightly weakens during EP El Niño events compared with CP events, with the center remaining close to the seasonal mean state location during JJA and shifting slightly east during SON (Figures 3g-3j). However, during EP El Niño in JJA, easterly winds flowing across West Antarctica are sourced from the continent, not the ASL (Figures S13b and S13d in Supporting Information S1). This results in a reduced impact on Antarctic precipitation and SMB gain as these winds are not moisture-laden (Figures 3g and 3i).



Geophysical Research Letters



Figure 3. Regression (a, c, d, g, h) and composite (b, e, f, i, j) SMB (from RACM02.3p3) and Antarctic atmospheric circulation (from ERA5) during CP JJA El Niño (a–c, e); EP JJA El Niño (g, i); CP SON El Niño (d, f); and EP SON El Niño (h, j). Pink (negative)-green (positive) shading indicates SMB regression (a, c, d, g, h) and composite (b, e, f, i, j) anomalies relative to the 1979–2018 averages for these variables. Purple (negative)-orange (positive) shading indicates MSLP regression (a, c, d, g, h) and composite (e, f, i, j) anomalies at 0.25 hPa increments. (b) Blue-red shading indicates average JJA total MSLP (hPa; from ERA5) and gray lines (dashed) indicate composite total MSLP for CP JJA El Niño above (and below) 1000 hPa. (e, f, i, j) Average seasonal composite "mean state" MSLP above 1000 hPa (gray lines, dashed below 1000 hPa). Red line indicates location of ASL center during (e) CP JJA, (f) CP SON, (i) EP JJA, and (j) EP SON. Composite MSLP anomalies are calculated by subtracting the average seasonal "mean state" MSLP (composite of all years 1979–2018) (gray lines) from the total MSLP (red line) composite during (e) CP JJA, (f) CP SON, (i) EP JJA, and (j) EP SON. All results shown are statistically significant at the 5% confidence level using a two-tailed Students' *t* test.

4. Discussion and Conclusions

We compare the regional differences in RACMO SMB (during JJA and SON) in response to CP and EP El Niño over the period 1979–2018. CP and EP El Niño have distinctly different impacts due to changes in teleconnections that lead to shifts in the position and strength of the ASL, and consequent changes in regions influenced by these El Niño types (Chen et al., 2023; H.-J. Lee & Jin, 2021; C. Zhang et al., 2021; Figure 3).



a) Regions in JJA with both CP and EP El Niño SMB Response



b)Regions in SON with both CP and EP El Niño SMB Response





Figure 4. Impact of different El Niño types on SMB in JJA (a) and SON (b). Regions which respond to both El Niño types with SMB increase (green), SMB decrease (pink), limited response (white; i.e., $-1 \le x \le 1 \text{ kg/m}^2$). Regions with opposing SMB responses (orange (CP increase, EP decrease) and blue (CP decrease, EP increase)). See Figures S14 and S15 in Supporting Information S1 for regions which respond to one El Niño type. Regions that are not statistically significant at the 5% confidence level using a two-tailed Students' *t* test are shown with 65% transparency.

We summarize the different impacts of CP and EP El Niño in JJA and SON on SMB in Figure 4 (Figures S14 and S15 in Supporting Information S1 for regions that only show resonses to one El Niño type). The largest impacts of EP and CP-related El Niño events on Antarctic SMB are evident in coastal zones and large parts of West Antarctica (Figure 4, Figure S14 in Supporting Information S1); however, there is significant variability in SMB in most coastal regions, which leads to a different magnitude and sign of SMB changes depending on the El Niño type and season. This variability is associated with changes in circulation, and upslope and downslope winds during the different ENSO types (Figures 1 and 3, Figure S13 in Supporting Information S1). Several regions—including parts of the Ross West and Dronning Maud Land regions and Wilkes Subglacial Basin—show strong SMB responses solely during one El Niño type (Figures S14b–S14c, S15b–S15c in Supporting Information S1).

Previous analyses of the climatic response of Antarctica to ENSO forcing include impacts on southern high latitude atmospheric circulation (Ciasto et al., 2015; J.-Y. Lee et al., 2021; Sun et al., 2013; Wilson et al., 2014). Our findings are consistent with these studies, with changes in the strength and location of the ASL being well-established (Chen et al., 2023; Clem et al., 2017; H.-J. Lee & Jin, 2021; C. Zhang et al., 2021). The precipitation and accumulation changes we identify here are also in agreement with other studies which identify ENSO-forced Antarctic precipitation changes, driven by anomalous circulation bringing warm, moisture-laden air over the continent (Chen et al., 2023; Cullather et al., 1996; Paolo et al., 2018; Sasgen et al., 2010). Only some of these studies consider how different El Niño types influence Antarctic surface climate (specifically temperature changes associated with large-scale circulation); however, none of these studies consider how different El Niño types influence et al., 1996; Paolo et al., 2018; Sasgen et al., 2018; Sasgen et al., 2010). EP El Niño tropical SST anomalies are generally larger than CP El Niño tropical SST anomalies (Figures S1d and S1e in Supporting Information S1), which has led to previous studies conventionally considering the EP El Niño associated atmospheric response to be stronger than CP (Chen et al., 2023; Sun et al., 2013; Wilson et al., 2014). By contrast, our findings show CP El Niño in JJA induces larger Antarctic SMB changes than EP El Niño (Figures 2a–2f) and CP and EP El Niño in SON induce similar Antarctic SMB changes (Figures 2g–2l).

Our findings are consistent with previous analyses of the impacts of ENSO forcing on Antarctic temperature (Clem et al., 2018; Ding & Steig, 2013; Li et al., 2014, 2015, 2021; B. Zhang et al., 2021). Previous studies have demonstrated that temperature decreases in the Antarctic Peninsula and increases in the East Antarctic and western Ross region are associated with El Niño (e.g., Li et al. (2014, 2015, 2021)). Our findings generally support those summarized in Li et al. (2021). However, by isolating CP and EP El Niño-driven impacts on temperature, our results suggest that the temperature changes synthesized in Li et al. (2021) align more closely to CP El Niño-driven changes rather than EP El Niño-driven changes. We also identify larger magnitude temperature anomalies during CP El Niño in JJA from regression analysis (not all significant) than from composite results (significant) (Figure 2a, Figure S5 in Supporting Information S1). In other words, previous studies (Kaitheri et al., 2021; King et al., 2023; Mémin et al., 2015; Paolo et al., 2018; M. Sasgen et al., 2010) may be underestimating the impact of El Niño on Antarctic SMB; regions may be exhibiting opposing responses to each El Niño type, effectively canceling out some of the impacts when ENSO diversity is not accounted for (Figure 4). Our results make clear the need to consider spatial variability in El Niño type to fully understand Antarctic climate impacts (van Dalum et al., 2022; Figure 3). For example, the integrated SMB change over the Ross West sector is approximately 4 kg/m² during both CP and EP ENSO events (Figure 3), which represents 10% of the seasonal SMB change in this region (van Dalum et al., 2021).

Our analysis is limited by a short data window (Text S8 in Supporting Information S1). The 40-year analysis period of this study (1979–2018) is the temporal extent of the RACMO2.3p3 data set. This 40-year record is not long enough to investigate decadal signals (Figure S4 in Supporting Information S1), therefore we focus on interannual variability. However, this 40-year timeseries is unlikely to capture the full range of variability expected in ENSO, which has been estimated to require 100–200 years timeseries length (Stevenson et al., 2010; Wittenberg, 2009). Furthermore, the 40-year period that we analyze likely includes impacts from both natural climate variability and anthropogenic forcings, between which we are unable to distinguish (Fox-Kemper et al., 2021; Meredith et al., 2019). Therefore, our results might not be representative of the full range of Antarctic climate impacts associated with El Niño spatial variability. Future work utilizing climate models and palae-oclimate data have the potential to better explain the CP and EP El Niño relationships with SMB.

Our results show the fundamental role that climate variability plays in Antarctic SMB. We highlight the need to consider how modes of variability such as ENSO will change with climate change, and impact Antarctic SMB. Projections of future ENSO characteristics remain uncertain (Fox-Kemper et al., 2021). Some CMIP6 ensemble results suggest a consensus on ENSO changes, projecting increased intensity and frequency of El Niño events (Cai et al., 2014, 2021; Shin et al., 2022). If these projections and teleconnections hold, our results suggest that greater changes in Antarctic SMB could occur with both El Niño types (Figure 2). The Freund et al. (2019) reconstruction of CP El Niño shows an increase in frequency over the 20th Century compared to the previous 400 years. If this increased frequency of CP El Niño events continues, as Shin et al. (2022) projects, our results suggest there may be significant Antarctic SMB impacts (Figures 2 and 3). Therefore, understanding future projections of El Niño, including changes in frequency, intensity, and specifically El Niño type is crucial for accurately predicting Antarctic SMB changes.

Data Availability Statement

All data sets are freely available. Regional Atmospheric Climate Model version 2.3 polar version 3 2-m temperature, precipitation, surface mass balance and melt are available at van Dalum et al. (2021) (https://doi.org/10. 5281/zenodo.7639053). ERA5 reanalyses data including mean sea level pressure and 10-m wind vectors are available at Hersbach et al. (2020) (https://doi.org/10.1002/qj.3803). El Niño indices, Niño-3 and Niño-4 are available at Rayner (2003) (https://doi.org/10.1029/2002JD002670).

References

- Agosta, C., Favier, V., Krinner, G., Gallée, H., Fettweis, X., & Genthon, C. (2013). High-resolution modelling of the Antarctic surface mass balance, application for the twentieth, twenty first and twenty second centuries. *Climate Dynamics*, 41(11–12), 3247–3260. https://doi.org/10. 1007/s00382-013-1903-9
- Ashok, K., Nakamura, H., & Yamagata, T. (2007). Impacts of ENSO and Indian Ocean dipole events on the Southern Hemisphere storm-track activity during austral winter. *Journal of Climate*, 20(13), 3147–3163. https://doi.org/10.1175/JCLI4155
- Bodart, J., & Bingham, R. (2019). The impact of the extreme 2015–2016 El Niño on the mass balance of the Antarctic ice sheet. *Geophysical Research Letters*, 46(23), 13862–13871. https://doi.org/10.1029/2019GL084466
- Bromwich, D., Guo, Z., Bai, L., & Chen, Q. (2004). Modeled Antarctic precipitation. Part I: Spatial and temporal variability. *Journal of Climate*, 17(3), 427–447. https://doi.org/10.1175/1520-0442(2004)017<0427:mappis>2.0.co;2
- Cai, W., Borlace, S., Lengaigne, M., van Rensch, P., Collins, M., Vecchi, G., et al. (2014). Increasing frequency of extreme El Niño events due to greenhouse warming. *Nature Climate Change*, 4(2), 111–116. https://doi.org/10.1038/nclimate2100
- Cai, W., Santoso, A., Collins, M., Dewitte, B., Karamperidou, C., Kug, J.-S., et al. (2021). Changing El Niño–Southern Oscillation in a warming climate. *Nature Reviews Earth and Environment*, 2(9), 628–644. https://doi.org/10.1038/s43017-021-00199-z
- Capotondi, A., Wittenberg, A. T., Newman, M., Di Lorenzo, E., Yu, J.-Y., Braconnot, P., et al. (2015). Understanding ENSO diversity. Bulletin of the American Meteorological Society, 96(6), 921–938. https://doi.org/10.1175/BAMS-D-13-00117.1
- Chen, X., Li, S., & Zhang, C. (2023). Distinct impacts of two kinds of El Niño on precipitation over the Antarctic Peninsula and West Antarctica in austral spring. Atmospheric and Oceanic Science Letters, 16(5), 100387. https://doi.org/10.1016/j.aosl.2023.100387
- Ciasto, L. M., Simpkins, G. R., & England, M. H. (2015). Teleconnections between tropical Pacific SST anomalies and extratropical Southern Hemisphere climate. Journal of Climate, 28(1), 56–65. https://doi.org/10.1175/JCLI-D-14-00438.1
- Clem, K. R., Bozkurt, D., Kennett, D., King, J. C., & Turner, J. (2022). Central tropical Pacific convection drives extreme high temperatures and surface melt on the Larsen C Ice Shelf, Antarctic Peninsula. *Nature Communications*, 13(1), 3906. https://doi.org/10.1038/s41467-022-31119-4
- Clem, K. R., Renwick, J. A., & McGregor, J. (2017). Large-scale forcing of the Amundsen Sea Low and its influence on sea ice and West Antarctic temperature. *Journal of Climate*, 30(20), 8405–8424. https://doi.org/10.1175/JCLI-D-16-0891.1
- Clem, K. R., Renwick, J. A., & McGregor, J. (2018). Autumn cooling of western East Antarctica linked to the tropical Pacific: ENSO and East Antarctic climate. *Journal of Geophysical Research: Atmospheres*, 123(1), 89–107. https://doi.org/10.1002/2017JD027435
- Cullather, R. I., Bromwich, D. H., & Van Woert, M. L. (1996). Interannual variations in Antarctic precipitation related to El Niño-Southern Oscillation. Journal of Geophysical Research, 101(D14), 19109–19118. https://doi.org/10.1029/96JD01769
- Dätwyler, C., Neukom, R., Abran, N. J., Gallant, A. J. E., Grosjean, M., Jacques-Coper, M., et al. (2018). Teleconnection stationarity, variability and trends of the Southern Annular Mode (SAM) during the last millennium. *Climate Dynamics*, 51(5–6), 2321–2339. https://doi.org/10.1007/ s00382-017-4015-0
- Diener, T., Sasgen, I., Agosta, C., Fürst, J. J., Braun, M. H., Konrad, H., & Fettweis, X. (2021). Acceleration of dynamic ice loss in Antarctica from satellite gravimetry. *Frontiers in Earth Science*, 9, 741789. https://doi.org/10.3389/feart.2021.741789
- Ding, Q., & Steig, E. J. (2013). Temperature change on the Antarctic Peninsula linked to the tropical Pacific. Journal of Climate, 26(19), 7570–7585. https://doi.org/10.1175/JCLI-D-12-00729.1
- Fogt, R. L., & Marshall, G. J. (2020). The southern annular mode: Variability, trends, and climate impacts across the Southern Hemisphere. WIREs Climate Change, 11(4). https://doi.org/10.1002/wcc.652
- Fox-Kemper, B., Hewitt, H., Xiao, C., AÃřalgeirsdóttir, G., Drifhout, S., Edwards, T., et al. (2021). Ocean, cryosphere and sea level change. In V. Masson-Delmotte, P. Zhai, A. Pirani, S. L. Connors, C. Péan, S. Berger, et al. (Eds.), *IPCC climate change 2021: The physical science basis. Contribution of working group 1 to the sixth assessment report of the Intergovernmental Panel on Climate Change*. Cambridge University Press. In Press.
- Freund, M. B., Henley, B. J., Karoly, D. J., McGregor, H. V., Abram, N. J., & Dommenget, D. (2019). Higher frequency of Central Pacific El Niño events in recent decades relative to past centuries. *Nature Geoscience*, 12(6), 450–455. https://doi.org/10.1038/s41561-019-0353-3
- Frieler, K., Clark, P. U., He, F., Buizert, C., Reese, R., Ligtenberg, S. R. M., et al. (2015). Consistent evidence of increasing Antarctic accumulation with warming. *Nature Change*, 5(4), 348–352. https://doi.org/10.1038/nclimate2574
- Guo, Z., Bromwich, D., & Hines, K. (2004). Modeled Antarctic precipitation. Part II: ENSO modulation over west Antarctica. *Journal of Climate*, *17*(3), 448–465. https://doi.org/10.1175/1520-0442(2004)017(0448:MAPPIE)2.0.CO;2
- Hanna, E., Pattyn, F., Navarro, F., Favier, V., Goelzer, H., Van Den Broeke, M. R., et al. (2020). Mass balance of the ice sheets and glaciers— Progress since AR5 and challenges. *Earth-Science Reviews*, 201, 102976. https://doi.org/10.1016/j.earscirev.2019.102976
- Hanna, E., Topál, D., Box, J. E., Buzzard, S., Christie, F. D. W., Hvidberg, C., et al. (2024). Short- and long-term variability of the Antarctic and Greenland ice sheets. *Nature Reviews Earth and Environment*, 5(3), 193–210. https://doi.org/10.1038/s43017-023-00509-7
- Hersbach, H., Bell, B., Berrisford, P., Hirahara, S., Horányi, A., Muñoz-Sabater, J., et al. (2020). The ERA5 global reanalysis [Dataset]. *Quarterly Journal of the Royal Meteorological Society*, 146(730), 1999–2049. https://doi.org/10.1002/qj.3803
- Jin, D., & Kirtman, B. P. (2009). Why the Southern Hemisphere ENSO responses lead ENSO. *Journal of Geophysical Research*, 114(D23), D23101. https://doi.org/10.1029/2009JD012657
- Kaitheri, A., Mémin, A., & Rémy, F. (2021). Inter-annual variability in the Antarctic ice sheets using geodetic observations and a climate model. *Remote Sensing*, 13(11), 2199. https://doi.org/10.3390/rs13112199
- Karamperidou, C., Stuecker, M. F., Timmermann, A., Yun, K., Lee, S., Jin, F., et al. (2020). ENSO in a changing climate: Challenges, paleoperspectives, and outlook. In M. J. McPhaden, A. Santoso, & W. Cai (Eds.), *Geophysical monograph series* (1st ed., pp. 471–484). Wiley. https://doi.org/10.1002/9781119548164.ch21

Acknowledgments

AM, ANM, FSM, AP, HVM and BJH were supported by the Australian Research Council Special Research Initiative for Securing Antarctica's Environmental Future (SR200100005). JMAM was supported by the Monash Graduate Scholarship (MGS) and Monash International Tuition Scholarship (MITS). FSM was supported under an Australian Research Council Discovery Early Career Research Award (DE210101433), CTVD was supported under Horizon 2020 (PROTECT grant no.869304). Open access publishing facilitated by Monash University, as part of the Wiley - Monash University agreement via the Council of Australian University Librarians.

- Kennicutt, M. C., Chown, S., Cassano, J., Liggett, D., Peck, L., Massom, R., et al. (2015). A roadmap for Antarctic and Southern Ocean science for the next two decades and beyond. *Antarctic Science*, 27(1), 3–18. https://doi.org/10.1017/S0954102014000674
- Kennicutt, M. C., Chown, S. L., Cassano, J. J., Liggett, D., Massom, R., Peck, L. S., et al. (2014). Polar research: Six priorities for Antarctic science. *Nature*, 512(7512), 23–25. https://doi.org/10.1038/512023a
- King, J., & Turner, J. (1997). Antarctic meteorology and climatology. Cambridge University Press.
- King, M., Lyu, K., & Zhang, X. (2023). Climate variability a key driver of recent Antarctic ice-mass change. Nature Geoscience, 16, 1128–1135. https://doi.org/10.1038/s41561-023-01317-w
- Kittel, C., Amory, C., Agosta, C., Jourdain, N. C., Hofer, S., Delhasse, A., et al. (2021). Diverging future surface mass balance between the Antarctic ice shelves and grounded ice sheet. *The Cryosphere*, 15(3), 1215–1236. https://doi.org/10.5194/tc-15-1215-2021
- Kug, J.-S., Jin, F.-F., & An, S.-I. (2009). Two types of El Niño events: Cold tongue El Niño and warm pool El Niño. Journal of Climate, 22(6), 1499–1515. https://doi.org/10.1175/2008JCLI2624.1
- Lee, H.-J., & Jin, E.-K. (2021). Seasonality and dynamics of atmospheric teleconnection from the tropical Indian Ocean and the western Pacific to west Antarctica. *Atmosphere*, *12*(7), 849. https://doi.org/10.3390/atmos12070849
- Lee, J.-Y., Marotzke, J., Bala, G., Cao, L., Corti, S., Dunne, J., et al. (2021). Future global climate: Scenario-based projections and NearTerm Information. In V. Masson-Delmotte, P. Zhai, A. Pirani, S. L. Connors, C. Péan, S. Berger, et al. (eds.), *IPCC climate change 2021: The physical science basis. Contribution of working group 1 to the sixth assessment report of the Intergovernmental Panel on Climate Change.* Cambridge University Press. In Press.
- Lee, T., Hobbs, W. R., Willis, J. K., Halkides, D., Fukumori, I., Armstrong, E. M., et al. (2010). Record warming in the South Pacific and western Antarctica associated with the strong central-Pacific El Niño in 2009–10: Record warming in the south Pacific. *Geophysical Research Letters*, 37(19). https://doi.org/10.1029/2010GL044865
- Li, X., Cai, W., Meehl, G. A., Chen, D., Yuan, X., Raphael, M., et al. (2021). Tropical teleconnection impacts on Antarctic climate changes. *Nature Reviews Earth and Environment*, 2(10), 680–698. https://doi.org/10.1038/s43017-021-00204-5
- Li, X., Holland, D. M., Gerber, E. P., & Yoo, C. (2014). Impacts of the north and tropical Atlantic Ocean on the Antarctic Peninsula and sea ice. *Nature*, 505(7484), 538–542. https://doi.org/10.1038/nature12945
- Li, X., Holland, D. M., Gerber, E. P., & Yoo, C. (2015). Rossby waves mediate impacts of tropical oceans on west Antarctic atmospheric circulation in austral winter. *Journal of Climate*, 28(20), 8151–8164. https://doi.org/10.1175/JCLI-D-15-0113.1
- Liu, H., Jezek, K., Li, B., & Zhao, Z. (2001). Radarsat Antarctic mapping project digital elevation model, version 2. NASA National Snow and Ice Data Center Distributed Active Archive Center. https://doi.org/10.5067/8JKNEW6BFRVD
- Medley, B., & Thomas, E. R. (2019). Increased snowfall over the Antarctic ice sheet mitigated twentieth-century sea-level rise. *Nature Climate Change*, 9(1), 34–39. https://doi.org/10.1038/s41558-018-0356-x
- Mémin, A., Flament, T., Alizier, B., Watson, C., & Rémy, F. (2015). Interannual variation of the Antarctic ice sheet from a combined analysis of satellite gravimetry and altimetry data. *Earth and Planetary Science Letters*, 422, 150–156. https://doi.org/10.1016/j.epsl.2015.03.045
- Meredith, M., Sommerkorn, M., Cassotta, S., Derksen, C., Ekaykin, A., Hollowed, A., et al. (2019). Polar regions. In H.-O. Pörtner, D. C. Roberts, V. Masson-Delmotte, P. Zhai, M. Tignor, E. Poloczanska, et al. (Eds.), *IPCC special report on the ocean and cryosphere in a changing climate*. In Press.
- Mottram, R., Hansen, N., Kittel, C., van Wessem, J. M., Agosta, C., Amory, C., et al. (2021). What is the surface mass balance of Antarctica? An intercomparison of regional climate model estimates. *The Cryosphere*, *15*(8), 3751–3784. https://doi.org/10.5194/tc-15-3751-2021
- Mudelsee, M. (2010). In *Climate time series analysis: Classical statistical and bootstrap methods* (Vol. 42). Springer. (OCLC: ocn646113953). Noël, B., Van Wessem, J. M., Wouters, B., Trusel, L., Lhermitte, S., & Van Den Broeke, M. R. (2023). Higher Antarctic ice sheet accumulation
- and surface melt rates revealed at 2 km resolution. *Nature Communications*, 14(1), 7949. https://doi.org/10.1038/s41467-023-43584-6 Paolo, F. S., Padman, L., Fricker, H. A., Adusumilli, S., Howard, S., & Siegfried, M. R. (2018). Response of Pacific-sector Antarctic ice shelves to
- the El Niño/Southern Oscillation. *Nature Geoscience*, 11(2), 121–126. https://doi.org/10.1038/s41561-017-0033-0 Pfeffer, J., Cazenave, A., & Barnoud, A. (2022). Analysis of the interannual variability in satellite gravity solutions: Detection of climate modes
- fingerprints in water mass displacements across continents and oceans. *Climate Dynamics*, 58(3–4), 1065–1084. https://doi.org/10.1007/ s00382-021-05953-z
- Rayner, N. A., Parker, D. E., Horton, E. B., Folland, C. K., Alexander, L. V., Rowell, D. P., et al. (2003). Global analyses of sea surface temperature, sea ice, and night marine air temperature since the late nineteenth century [Dataset]. *Journal of Geophysical Research*, 108(D14), 4407. https://doi.org/10.1029/2002JD002670
- Ren, H.-L., & Jin, F.-F. (2011). Niño indices for two types of ENSO: Niño indices for two types of ENSO. Geophysical Research Letters, 38(4). https://doi.org/10.1029/2010GL046031
- Renwick, J. A., & Revell, M. J. (1999). Blocking over the South Pacific and Rossby Wave Propagation. *Monthly Weather Review*, 127(10), 2233–2247. https://doi.org/10.1175/1520-0493(1999)127(2233:BOTSPA)2.0.CO;2
- Ribera, P., & Mann, M. E. (2003). ENSO related variability in the Southern Hemisphere, 1948-2000: ENSO related variability in the Southern Hemisphere. *Geophysical Research Letters*, 30(1), 6-1–6–4. https://doi.org/10.1029/2002GL015818
- Rignot, E., Mouginot, J., & Scheuchl, B. (2011). Ice flow of the Antarctic ice sheet. Science, 333(6048), 1427–1430. https://doi.org/10.1126/ science.1208336
- Rignot, E., Velicogna, I., van den Broeke, M. R., Monaghan, A., & Lenaerts, J. T. M. (2011). Acceleration of the contribution of the Greenland and Antarctic ice sheets to sea level rise: Acceleration of Ice Sheet loss. *Geophysical Research Letters*, 38(5). https://doi.org/10.1029/ 2011GL046583
- Sasgen, I., Dobslaw, H., Martinec, Z., & Thomas, M. (2010). Satellite gravimetry observation of Antarctic snow accumulation related to ENSO. Earth and Planetary Science Letters, 299(3–4), 352–358. https://doi.org/10.1016/j.epsl.2010.09.015
- Saunderson, D., Mackintosh, A. N., McCormack, F. S., Jones, R. S., & Van Dalum, C. T. (2024). How does the southern annular mode control surface melt in East Antarctica? *Geophysical Research Letters*, 51(6), e2023GL105475. https://doi.org/10.1029/2023GL105475
- Seroussi, H., Nowicki, S., Payne, A. J., Goelzer, H., Lipscomb, W. H., Abe-Ouchi, A., et al. (2020). ISMIP6 Antarctica: A multi-model ensemble of the Antarctic ice sheet evolution over the 21st century. *The Cryosphere*, 14(9), 3033–3070. https://doi.org/10.5194/tc-14-3033-2020
- Shin, N.-Y., Kug, J.-S., Stuecker, M. F., Jin, F.-F., Timmermann, A., & Kim, G.-I. (2022). More frequent central Pacific El Niño and stronger eastern Pacific El Niño in a warmer climate. npj Climate and Atmospheric Science, 5(1), 101. https://doi.org/10.1038/s41612-022-00324-9
- Song, H.-J., Choi, E., Lim, G.-H., Kim, Y. H., Kug, J.-S., & Yeh, S.-W. (2011). The central Pacific as the export region of the El Niño-Southern Oscillation sea surface temperature anomaly to Antarctic sea ice: ENSO SST to ADP though Central Pacific. *Journal of Geophysical Research*, 116(D21). https://doi.org/10.1029/2011JD015645
- Stevenson, S., Fox-Kemper, B., Jochum, M., Rajagopalan, B., & Yeager, S. G. (2010). ENSO model validation using wavelet probability analysis. Journal of Climate, 23(20), 5540–5547. https://doi.org/10.1175/2010JCL13609.1



- Sun, D., Xue, F., & Zhou, T. (2013). Impacts of two types of El Niño on atmospheric circulation in the Southern Hemisphere. Advances in Atmospheric Sciences, 30(6), 1732–1742. https://doi.org/10.1007/s00376-013-2287-9
- Turner, J. (2004). The El Niño-southern oscillation and Antarctica: ENSO and the Antarctic. International Journal of Climatology, 24(1), 1–31. https://doi.org/10.1002/joc.965
- van Dalum, C. T., van de Berg, W. J., & van den Broeke, M. (2021). RACMO2.3p3 monthly SMB, SEB and t2m data for Antarctica (1979-2018) [Dataset]. Zenodo. https://doi.org/10.5281/ZENOD0.5512076
- van Dalum, C. T., van de Berg, W. J., & van den Broeke, M. R. (2022). Sensitivity of Antarctic surface climate to a new spectral snow albedo and radiative transfer scheme in RACMO2.3p3. *The Cryosphere*, *16*(3), 1071–1089. https://doi.org/10.5194/tc-16-1071-2022
- Wilson, A. B., Bromwich, D. H., Hines, K. M., & Wang, S.-H. (2014). El Niño flavors and their simulated impacts on atmospheric circulation in the high southern latitudes. *Journal of Climate*, 27(23), 8934–8955. https://doi.org/10.1175/JCLI-D-14-00296.1
- Wittenberg, A. T. (2009). Are historical records sufficient to constrain ENSO simulations? *Geophysical Research Letters*, 36(12), L12702. https://doi.org/10.1029/2009GL038710
- Yiu, Y. Y. S., & Maycock, A. C. (2020). The linearity of the El Niño teleconnection to the Amundsen Sea region. Quarterly Journal of the Royal Meteorological Society, 146(728), 1169–1183. https://doi.org/10.1002/qj.3731
- Zhan, J., Shi, H., Wang, Y., & Yao, Y. (2021). Complex principal component analysis of Antarctic ice sheet mass balance. *Remote Sensing*, 13(3), 480. https://doi.org/10.3390/rs13030480
- Zhang, B., Yao, Y., Liu, L., & Yang, Y. (2021). Interannual ice mass variations over the Antarctic ice sheet from 2003 to 2017 were linked to El Niño-Southern Oscillation. *Earth and Planetary Science Letters*, 560, 116796. https://doi.org/10.1016/j.epsl.2021.116796
- Zhang, C., Li, T., & Li, S. (2021). Impacts of CP- and EP-El Niño events on the Antarctic sea ice in austral spring. *Journal of Climate*, 1–76. https://doi.org/10.1175/JCLI-D-21-0002.1